UC Berkeley UC Berkeley Previously Published Works

Title

Temporal Trends of Racial and Socioeconomic Disparities in Population Exposures to Upstream Oil and Gas Development in California

Permalink https://escholarship.org/uc/item/5mf2c8vb

Journal GeoHealth, 7(3)

ISSN 2471-1403

Authors

González, David JX Morton, Claire M Hill, Lee Ann L <u>et al.</u>

Publication Date 2023-03-01

DOI

10.1029/2022gh000690

Peer reviewed



GeoHealth

RESEARCH ARTICLE

10.1029/2022GH000690

Key Points:

- Black, Hispanic, and socioeconomically marginalized people persistently had disproportionately high exposure to oil and gas development
- The widest observed disparities were for Black people residing in neighborhoods with the most intensive oil and gas production
- Disparities in exposure to oil and gas development may contribute to previously reported health disparities

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. J. X. González, djxgonz@berkeley.edu

Citation:

González, D. J. X., Morton, C. M., Hill, L. A. L., Michanowicz, D. R., Rossi, R. J., Shonkoff, S. B. C., et al. (2023). Temporal trends of racial and socioeconomic disparities in population exposures to upstream oil and gas development in California. *GeoHealth*, 7, e2022GH000690. https://doi. org/10.1029/2022GH000690

Received 4 AUG 2022 Accepted 16 FEB 2023

Author Contributions:

Conceptualization: David J. X. González, Drew R. Michanowicz, Joan A. Casey, Rachel Morello-Frosch Data curation: David J. X. González, Claire M. Morton, Drew R. Michanowicz Formal analysis: David J. X. González Funding acquisition: Joan A. Casey, Rachel Morello-Frosch Investigation: David J. X. González, Claire M. Morton

© 2023 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Temporal Trends of Racial and Socioeconomic Disparities in Population Exposures to Upstream Oil and Gas Development in California

David J. X. González^{1,2}, Claire M. Morton³, Lee Ann L. Hill⁴, Drew R. Michanowicz⁴, Robert J. Rossi⁴, Seth B. C. Shonkoff^{1,4,5}, Joan A. Casey^{6,7}, and Rachel Morello-Frosch^{1,2}

¹Division of Environmental Health Sciences, School of Public Health, University of California, Berkeley, Berkeley, CA, USA, ²Department of Environmental Science, Policy, and Management, University of California, Berkeley, Berkeley, CA, USA, ³Mathematical and Computational Science Program, Stanford University, Stanford, CA, USA, ⁴PSE Healthy Energy, Oakland, CA, USA, ⁵Lawrence Berkeley National Laboratory, Energy Technologies Area, Berkeley, CA, USA, ⁶Department of Environmental Health Sciences, Columbia University, New York, NY, USA, ⁷Department of Environmental and Occupational Health Sciences, University of Washington, Seattle, WA, USA

Abstract People living near oil and gas development are exposed to multiple environmental stressors that pose health risks. Some studies suggest these risks are higher for racially and socioeconomically marginalized people, which may be partly attributable to disparities in exposures. We examined whether racially and socioeconomically marginalized people in California are disproportionately exposed to oil and gas wells and associated hazards. We longitudinally assessed exposure to wells during three time periods (2005–2009, 2010–2014, and 2015–2019) using sociodemographic data at the census block group-level. For each block group and time period, we assessed exposure to new, active, retired, and plugged wells, and cumulative production volume. We calculated risk ratios to determine whether marginalized people disproportionately resided near wells (within 1 km). Averaged across the three time periods, we estimated that 1.1 million Californians (3.0%) lived within 1 km of active wells. Nearly 9 million Californians (22.9%) lived within 1 km of plugged wells. The proportion of Black residents near active wells was 42%-49% higher than the proportion of Black residents across California, and the proportion of Hispanic residents near active wells was 4%-13% higher than their statewide proportion. Disparities were greatest in areas with the highest oil and gas production, where the proportion of Black residents was 105%-139% higher than statewide. Socioeconomically marginalized residents also had disproportionately high exposure to wells. Though oil and gas production has declined in California, marginalized communities persistently had disproportionately high exposure to wells, potentially contributing to health disparities.

Plain Language Summary People living near oil and gas wells are exposed to pollutants that may adversely affect their health. We investigated whether racially or socioeconomically marginalized people in California had disproportionately higher exposure to wells. We also assessed whether disparities were wider in areas with more intensive oil and gas production. We examined changes in neighborhood-level sociodemographic characteristics in three time periods: 2005–2009, 2010–2014, and 2015–2019. For each neighborhood and time period, we estimated the number of wells and the total volume of oil and gas produced within 1 km (0.62 miles). We estimated that approximately 1.1 million Californians (3.0%) lived within 1 km of active wells between 2005 and 2019. The proportion of Black residents living near active wells was 42%–49% higher than the proportion of Black residents across California, and the proportion of Latinx residents near active wells was 4%–13% higher than their statewide proportion. For Black people, the disparities were widest in Los Angeles County neighborhoods with the most intensive production. Socioeconomically marginalized people also had disproportionately high exposure. Most disparities persisted throughout the 15-year study period. Black, Hispanic, and socioeconomically marginalized people had disproportionately high exposure to wells, potentially contributing to health disparities.

1. Introduction

Oil and gas development near residences increases risks of exposure to water pollution, air pollution, noise, and other stressors that may adversely affect health and socioeconomic well-being (Adgate et al., 2014; Allshouse et al., 2019; DiGiulio & Shonkoff, 2021; Garcia-Gonzales, Shamasunder, & Jerrett, 2019; Garcia-Gonzales,



Methodology: David J. X. González, Claire M. Morton, Lee Ann L. Hill, Drew R. Michanowicz, Robert J. Rossi, Seth B. C. Shonkoff, Joan A. Casey, Rachel Morello-Frosch

Project Administration: David J. X. González, Joan A. Casey, Rachel Morello-Frosch Software: David J. X. González, Claire M. Morton Supervision: Joan A. Casev, Rachel Morello-Frosch Validation: David J. X. González Visualization: David J. X. González, Claire M. Morton Writing - original draft: David J. X. González, Claire M. Morton Writing - review & editing: David J. X. González, Lee Ann L. Hill, Drew R. Michanowicz, Robert J. Rossi, Seth B. C. Shonkoff, Joan A. Casev, Rachel Morello-Frosch

Shonkoff, et al., 2019; Gonzalez, Francis, et al., 2022; Hays et al., 2017; McKenzie et al., 2018; Shonkoff, Hill, & Domen, 2021; Shonkoff, Jordan, et al., 2015; Shonkoff, Maddalena, et al., 2015; Williams et al., 2021). Studies from California have reported that residing near new and active oil and gas wells is associated with higher risk of adverse reproductive and respiratory health outcomes (Gonzalez et al., 2020; Johnston et al., 2021; Shamasunder et al., 2018; Tran et al., 2020, 2021). Indeed, the California Oil and Gas Public Health Rulemaking Scientific Advisory Panel applied the Bradford Hill Criteria for Causation and concluded with a high degree of certainty that there was a causal association between living near oil and gas development and adverse respiratory and perinatal outcomes (Hill, 1965; Lucas & McMichael, 2005; Shonkoff, Morello-Frosch, et al., 2021). Plugged and abandoned wells may also be harmful to health, due to emissions of non-methane volatile organic compounds (Solis, 2022; Williams et al., 2021). Several studies have reported worse health outcomes among racially and socioeconomically marginalized people living near oil and gas development, and, in several regions in North America, marginalized communities have disproportionately high exposure to wells and associated infrastructure (Caron-Beaudoin et al., 2018, 2019, 2022; Casey et al., 2019; Cushing et al., 2020; Gonzalez et al., 2020; Gonzalez, Nardone, et al., 2022; Kroepsch et al., 2019; Proville et al., 2022; Tran et al., 2020, 2021).

Observed health disparities may be attributable to disproportionately high exposure to chemical contaminants and other stressors from upstream oil and gas development, defined as activities that include preproduction (drilling new wells or re-working and maintaining existing wells), production (extraction of oil and gas from active wells, oil-water separation, hydrocarbon processing, etc.), and postproduction (retiring and plugging wells). Each stage of production is associated with environmental stressors that may plausibly contribute to adverse health outcomes. These stressors include emissions of air pollutants, groundwater contamination, persistent noise and odors, seismic activity, and social disruption (Adgate et al., 2014; Deziel, Clark, et al., 2022; Garcia-Gonzales, Shonkoff, et al., 2019). Assessing individual- or area-level exposure to wells based on proximity provides a holistic indicator for the aggregate hazards associated with oil and gas development (Deziel, Clark, et al., 2022).

Several studies have found that racially and socioeconomically marginalized populations have disproportionately high exposure to oil and gas wells in several parts of the United States (U.S.). Johnston et al. (2016) found that, in the Eagle Ford region in southern Texas, the proportion of people of color living near oil and gas disposal wells was significantly higher than the proportion of non-Hispanic white people. A follow-up study (Johnston et al., 2020) found that Hispanic residents had disproportionately high exposure to natural gas flaring. An analysis of wastewater injection wells in Ohio found that census block-group level median household income was inversely associated with the presence of wells (Silva et al., 2018). Similarly, cross-sectional analyses by the Natural Resources Defense Council and FracTracker Alliance report that racially and socioeconomically marginalized communities have disproportionately high exposure to oil and gas wells in Los Angeles and Kern Counties, California (Ferrar, 2020; Srebotnjak & Rotkin-Ellman, 2014). Disparities have also been observed in Canada, with higher exposures to VOCs and trace metals among Indigenous women living in British Columbia compared to non-Indigenous pregnant women (Caron-Beaudoin et al., 2018, 2019, 2022). Exposure disparities have been documented dating back to the early twentieth century, with evidence that these disparities are attributable, at least in part, to structural racist policies such as redlining and housing segregation (Cumming, 2018; Gonzalez, Nardone, et al., 2022; Quam-Wickham, 1998).

Despite widespread exposure to upstream oil and gas development, the extent of oil and gas-related exposure disparities among racially and socioeconomically marginalized populations in California remains poorly characterized (Czolowski et al., 2017; Michanowicz et al., 2019). To our knowledge, these disparities have only been examined cross-sectionally (i.e., at one point in time) and without accounting for the life cycle of oil and gas wells (from preproduction to active production to retirement).

In this study, we longitudinally examined sociodemographic characteristics of communities near upstream oil and gas wells across major stages of development in California. We had two overarching aims. First, we aimed to determine whether racially or socioeconomically marginalized populations in California had disproportionately high exposure to new, active, retired, and plugged wells. Among exposed populations, we also investigated the characteristics of communities with more intensive exposures as indicated by greater cumulative production volume (Gonzalez, Francis, et al., 2022). Second, we aimed to examine whether observed disparities were persistent throughout the study period beginning with the first set of data collected by the American Community Survey (ACS) in 2005 through 2019. Among communities that were exposed to any wells throughout the study period, we compared sociodemographic characteristics in communities where the intensity of well drilling and



the volume of production increased versus places where these activities decreased. In additional sets of analyses, we examined differences among the three counties with the greatest number of people living within 1 km of wells: Los Angeles, Orange, and Kern. Collectively, over 90% of Californians residing near new, active, and retired wells during the study period lived in these three counties. Findings from this study may help affected residents, researchers, and policymakers to better understand environmental justice-related concerns in the context of oil and gas development across California.

2. Methods

2.1. Study Design

To conduct our longitudinal assessment of disparities of exposure to upstream oil and gas development activities, we leveraged time-series data on sociodemographic characteristics and upstream oil and gas development from 2005 to 2019. We divided our study period into three time periods to align with data available from the ACS: 2005–2009, 2010–2014, and 2015–2019. Estimates based on the ongoing ACS are available for single years and for multiple years, with aggregation up to 5 years. Data in the 5-year ACS estimates are representative of the period of data collection and are considered more reliable for areas with small populations and for relatively small demographic groups (U.S. Census Bureau, 2022). For purposes of the current study, we began the study period in 2005 because this was the first year for which 5-year (2005–2009) ACS data are available. The geographic units of observation for these analyses were census block groups, which are contiguous areas nested within census tracts that typically contain 600 to 3,000 residents (U.S. Census Bureau, 2022). To investigate temporal variation in oil and gas activities, we examined trends in the drilling of new wells, production of oil and gas at active wells, and retirement of wells post-production. We also investigated proximity to wells that were plugged and abandoned prior to the study period.

2.2. Study Population

The study population comprised people who resided in California between 2005 and 2019. We used census block group-level sociodemographic data from the 5-year averaged ACS for three periods: 2005–2009, 2010–2014, and 2015–2019. With the added temporal component of these analyses, our unit of analysis was the block group-period. We used data from block group-periods when comparing aggregated state- and county-level sociodemographic data.

We were specifically interested in estimating exposure among residents with the following racial/ethnic identities: Hispanic/Latinx, non-Hispanic Asian, non-Hispanic Black, non-Hispanic white, American Indian or Alaska Natives, and those belonging to other marginalized racial/ethnic groups. For purposes of this study, we consider Hispanic/Latinx, non-Hispanic Asian, non-Hispanic Black, and American Indian and Alaska Natives as racially marginalized. Throughout the manuscript, we use the gender-inclusive term Hispanic and Latinx. We also assessed exposure to oil and gas development activities on land managed by federally- or state-recognized tribes. For this set of analyses, we used the American Indian/Alaska Native/Native Hawaiian Area National Shapefile provided by the U.S. Census Bureau, which includes the geographic extent of federally- and state-recognized American Indian reservations as well as lands held in trust (U.S. Census Bureau, 2018).

We recognize that there is broad heterogeneity among people within each racial/ethnic group representing diverse histories and experiences of marginalization. However, disaggregated data on race/ethnicity and tribal affiliation were not available from ACS. In California, there are 109 federally recognized Indian tribes and 45 tribal communities that were formerly recognized or that have never been recognized by the federal government (California Tribal Court-State Court Forum, n.d.). Tribal communities are located throughout the state, including in approximately 100 reservations or Rancherías as well as in urban settings. Los Angeles and San Francisco have some of the largest urban Indigenous populations in the U.S., including approximately 70,000 people with tribal origins in other states, many of whom are descendants of people who were forcefully relocated due to federal policy (California Tribal Court-State Court Forum, n.d.). Among the Hispanic and Latinx population of California, in 2010, the majority (81.5%) were of Mexican descent, 4.1% were of Salvadoran descent, 2.4% were of Guatemalan descent, and 12.0% had familial origins in other countries (Ennis et al., 2011). Among the state's Asian population in 2010, 26.6% were of Filipino descent, 26.1% were of Chinese descent, 11.7% were of Japanese descent (Hoeffel





Figure 1. (a) Map of California showing 1, 3, and 10 km buffers around all wells in the preproduction, production, and postproduction stages during the study period, 2005–2019. (b) Midway-Sunset first purchase price of a barrel of oil in U.S. dollars by month, an indicator for California oil prices. (c) Annual count of wells in preproduction in California, including wells spudded or completed. (d) Sum of annual oil and gas production volume in hundreds of millions of barrels of oil equivalent at all active wells. (e) Wells entering the postproduction stage (i.e., retired) by year, including plugged and long-term idle wells (\geq 8 years without production).

et al., 2012). Additionally, several racially marginalized groups are subsumed under the non-Hispanic white classification, such as many people of North African or Middle Eastern descent.

Finally, we also assessed exposure among residents who were socioeconomically marginalized. Specifically, we examined the following five indicators: proportion of residents with income below 200% of the federal poverty level, proportion of adults aged 25 and older with fewer than 12 years of education, proportion of renters, linguistically isolated people (i.e., adults 18 years or older who do not speak English "very well"), and proportion of eligible voters who did not participate averaged across the 2012 and 2016 elections. The U.S. government established their definition of the poverty level in the 1960s and has provided adjustments based on the initial definition. However, to meet basic needs, households require at least two times the federal level, which is why we based our definition on 200% of the federal poverty level (Su et al., 2009). Voter turnout data from the first study period (2005–2009) were unavailable and consequently we omitted voter turnout from our analyses of that time period.

2.3. Exposure Assessment

We obtained data on oil and gas wells from the California Geologic Energy Management Division (CalGEM), the state agency with regulatory authority over well permitting and many other aspects of oil and gas development in the state, as well as Enverus, a private data aggregation service. We assembled a data set that, for each well, included the geographic coordinates, spud date (the initiation of drilling and beginning of preproduction), completion date (end of preproduction), first and last dates of production, abandonment date, and well type. Additionally, we obtained data on the cumulative annual volume of oil and gas produced at each well, measured in barrels of oil equivalent (BOE).

We assessed exposure to all new, active, and retired wells across California (Figure 1). We defined new wells as those in the preproduction stage, between spudding (initiation of drilling) and completion (when the well is ready to start production). Active wells were defined as those with >0 BOE of cumulative oil or gas produced in any year of the study period. Finally, we defined retired wells as those that entered the postproduction phase during the study period. We defined the postproduction phase as starting on the reported date of abandonment or, where no abandonment date was reported, 8 years after the last production date in accordance with the state's definition of long-term idle wells (California Code PRC § 3008). We also assessed exposure to all plugged wells in cross-section for the 2015-2019 time period, as we expected any temporal variation in exposure to plugged wells during the study period to correlate with exposure to well retirements, which we assessed separately. Specifically, we defined plugged wells as those that were plugged and abandoned at any time, including wells retired before the study period began in 2005. In the plugged wells assessment, we included the 44.3% of plugged wells with no reported last production or abandonment dates, which we assumed were plugged prior to 2019. Notably, wells may have been in multiple production stages throughout the study period. In cases

where a well was in two production stages during the same 5-year time period (e.g., in preproduction in 2005 and production in 2006), we included it in both the new and active wells exposure assessments.

For purposes of the current study, we defined exposed populations as those living within 1 km of oil and gas wells. In assessing exposure, we used a similar approach to proximity metrics commonly used in the environmental





Figure 2. Illustration of the two exposure assessment protocols we used to estimate exposure to wells at the census block group-level, with real data from Carson and Wilmington in Los Angeles County, predominantly Black and Latinx communities with intensive oil and gas production. Shown are a sample block group (highlighted in yellow) and all wells that were new, active, or retired during the study period, 2005–2019. (a) With the areal apportionment method our aim was to assess the proportion of residents who were exposed to wells in each stage of production, with the simplifying assumption that residents are evenly distributed throughout the block group. We assessed the proportion of each block group that intersected with 1 km buffers around new, active, or retired wells. In this example, all residents were exposed to new and active wells and approximately 10% were exposed to retired wells. (b) With the block group centroid method our aim was, among block groups with any exposure, to assess the intensity of the exposure, with the simplifying assumption that residents are in the center of the block group. We counted the number of wells in each development stage, and the sum of oil and gas production (barrels of oil equivalent), within 1 km of the block group centroid. In this example, neighborhood residents were exposed to 1 new well and 12 active wells (and the cumulative sum of oil and gas produced at those active wells).

epidemiology literature (Deziel, Clark, et al., 2022). We assessed exposure for each census block group and time period (2005-2009, 2010-2014, and 2015-2019) with the aim first of estimating populations exposed to any wells and second, among exposed block groups, of estimating the intensity of production in BOE. First, we used areal apportionment to estimate populations that had any exposure to wells (Figure 2a). To do this, we determined the overlap between each block group and a 1 km buffer around wells in each production stage. We then determined the area of the block group that overlapped with the 1 km wells buffer and assigned exposure to residents in proportion to the area of overlap, with the simplifying assumption that residents were evenly distributed within the block group. For example, if a block group had 1,000 residents and 30% of the block group area intersected the well buffer, we assumed that 300 residents were exposed. We also apportioned residents within each sociodemographic group of interest in proportion to the percent overlap between the well buffer and block group. If, for example, 50% of residents in that same block group were Latinx, we would say that 150 of the exposed residents were Latinx. We completed this procedure for each block group-period for wells in each stage of production. Second, among block groups with any exposure to wells, we ascertained the intensity of the exposure by taking the sum of wells or production volume within 1 km of the block group centroid (Figure 2b). Specifically, for each block group-period, we counted the number of new, active, and retired wells (and plugged wells for 2015–2019). Additionally, for active wells, we assessed the cumulative sum of oil and gas production volume in BOE during that time-period for all wells within 1 km of the block group centroid. As Deziel, Clark, et al. (2022) and Deziel, McKenzie, et al., 2022 have noted, proximity-based exposure metrics have the advantages of being appropriate for retrospective analyses, easily scalable, and effective for capturing aggregate exposures to chemical, physical, and other stressors associated with oil and gas production, though the approach is constrained in not being able to identify specific etiological agents or risks beyond the buffer distance.

2.4. Statistical Analyses

We calculated summary statistics for exposure among each of the sociodemographic groups of interest identified above for each of the three time periods. We derived risk ratios to evaluate whether marginalized populations had



disproportionately high exposure to wells. Specifically, to quantify the relative risk for each sociodemographic group of interest, we calculated a group risk ratio (RR):

Population_{exposed}}{\sum Population_{total}},

or the ratio of the proportion of individuals in a specific demographic group who were exposed to the proportion of the total population who were exposed. A RR of greater than 1 indicates that the group was over-represented among the exposed population. Conversely, groups that were under-represented among the exposed population would have a RR of less than 1. Groups that were exposed in proportion that group's total population would have a RR equal to 1. For example, in 2005–2009, we estimated that 100,941 of the 2,172,247 (4.64%) non-Hispanic Black residents of California resided within 1 km of an active well. We also estimated that 1,169,930 of all 36,308,527 (3.22%) California residents lived near active wells. Thus, the RR for this group was (100,941/2,172,247)/(1,169,930/36,308,527), or 1.44. (We could also have taken 4.64/3.22 to get to the same result). Since the RR was great than 1, we concluded that this demographic group had disproportionately high exposure in 2005–2009. Using this approach, we estimated the RR for each sociodemographic group of interest to determine whether they had disproportionately high or low exposure to oil and gas wells in each stage of production (new, active, retired, and plugged wells). We compared RRs through time among racial/ethnic groups and each of the socioeconomic indicators (people living in poverty, people with <12 years of educational attainment, renter-occupied households, linguistically isolated, and non-voters).

We also conducted these analyses at the county level for the three counties with the highest number of residents exposed to oil and gas development: Los Angeles, Orange, and Kern. For this set of analyses, we calculated RRs as described above, but with comparison to the county-level demographic composition rather than the state-level demographics.

In a secondary analysis, we analyzed the sociodemographic characteristics of block groups where there were changes in the count of new wells or sum of cumulative production volume between 2010–2014 and 2015–2019. We were interested in a panel data approach to investigating whether changes in intensity of oil and gas drilling and production were associated with the racial/ethnic composition of exposed block groups. Because the U.S. Census Bureau redefines block group boundaries at each decennial census, for this set of within-block group analyses we used data for the latter two time periods. We then calculated the differences for both exposure metrics (new wells, cumulative production volume) between the two periods and categorized block groups by whether they had an increase, decrease, or no change for either metric. Among block groups within each category, we then summed the number of people in each racial/ethnic group.

2.5. Ethical Considerations

Because we used publicly available data for our analysis, our study does not fall within the definition of human subjects research.

3. Results

3.1. Demographics

In the first of the three time periods considered in this study (2005–2009), California had an estimated population of 36,308,527 residing in 22,133 census block groups (Table 1). The statewide population grew by 8.2% to 39,283,497 in 23,212 block groups by the third time period (2015–2019). The percentage of residents who were Hispanic or Latinx increased from 36.1% in 2005–2009 to 39.0% in 2015–2019. Similarly, the non-Hispanic Asian population grew from 12.1% to 14.3% and the population that identified as two or more races increased from 2.1% to 3.0%. The proportion of the American Indian or Alaska Native population decreased slightly from 0.5% to 0.4%, the proportion of the non-Hispanic Black population decreased from 6.0% to 5.5%, and the proportion of the non-Hispanic white population decreased from 42.5% to 37.2%. The proportion of people living in poverty rose from 11.6% to 12.5%. With respect to educational attainment, the proportion of residents >25 years old who had not completed high school decreased from 19.5% in 2005–2009 to 16.7% in 2015–2019. The percent of renters increased from 42.1% to 45.2%. Between 20.0% and 21.9% of households were linguistically isolated during the study period. Across block groups, an average of 70.0 \pm 10.9% of eligible voters participated in the 2012 and 2016 elections.

Table 1

Summary of Sociodemographic Characteristics of California Residents During Each of the Three Time Periods Considered in the Current Study

	Time period		
	2005–2009	2010–2014	2015–2019
California population, <i>n</i>	36,308,527	38,066,920	39,283,497
Census block groups, n	22,133	23,212	23,212
Race/ethnicity, n (%)			
Hispanic or Latinx	13,102,161 (36.1)	14,534,449 (38.2)	15,327,688 (39.0)
Non-Hispanic nor Latinx			
American Indian or Alaska Native	164,754 (0.5)	145,736 (0.4)	140,831 (0.4)
Asian	4,410,395 (12.1)	5,062,736 (13.3)	5,610,931 (14.3)
Black	2,172,247 (6.0)	2,155,929 (5.7)	2,169,155 (5.5)
White	15,446,196 (42.5)	14,905,601 (39.2)	14,605,312 (37.2)
Other	126,774 (0.3)	81,869 (0.2)	100,119 (0.3)
Two or more races	762,608 (2.1)	1,044,136 (2.7)	1,188,673 (3.0)
Educational attainment, n (%)			
<12 years	4,537,564 (19.5)	4,602,986 (18.5)	4,418,675 (16.7)
12 years	5,078,536 (21.9)	5,153,257 (20.7)	5,423,462 (20.5)
>12 years	13,603,117 (58.6)	15,109,623 (60.8)	16,629,406 (62.8)
Living under 2x poverty line, n (%)	1,411,227 (11.6)	1,833,084 (14.5)	1,632,747 (12.5)
Renter-occupied households, n (%)	5,125,759 (42.1)	5,708,355 (45.2)	5,889,686 (45.2)
Linguistically isolated adults, n (%)	5,893,258 (21.9)	6,171,673 (21.4)	6,038,794 (20.0)
Voter turnout, n (%)	_	13,360,940 (71.6)	13,360,940 (71.6)

Note. Percentages for educational attainment and living in poverty determined from reported education and poverty universes, respectively. Household-level data for renters for the universe of California households. The percentage of linguistically isolated adults is calculated from the sum of total adult population of California during each time period. We reported average turnout and percent of voters among eligible residents using averaged data from the 2012 and 2016 elections; we applied these data in both the 2010–2014 and 2015–2019 periods.

3.2. Oil and Gas Wells

Across California, there were a total of 110,062 new, active, or retired wells between 2005 and 2019. There was substantial temporal variation in oil and gas operations. Some 30,713 wells were in preproduction during the study period, with drilling activity peaking in 2012 and decreasing through 2019 (Table S1 in Supporting Information S1). There were 83,583 active oil and gas wells in California during the study period, with production volume totaling 4.06 billion BOE. Among active wells, mean annual production was 24,754 \pm 2,489,562 BOE (range: 0.1; 310,118,585). Statewide annual oil and gas production decreased by 33.5% throughout the study period, from 310 million BOE in 2005 to 206 million BOE in 2019 (Figure 1d). Some 31,199 wells (28.3%) were retired or long-term idle (i.e., no production for at least 8 years) during the study period. Of these, 22,774 (73%) were plugged wells and 8,425 (27%) were long-term idle. Well retirements increased from 1,639 in 2005 to 3,503 in 2013, then decreased to 1,604 in 2019. There were 126,560 total plugged wells in the data set, including wells retired before the study period. Of these, 100,697 (79.6%) had a reported spud or completion date, which ranged from March 1882 to December 2019. Abandonment dates ranged from February 1887 to June 2018. The duration of production for plugged wells was 10.9 \pm 8.2 years, though duration data were missing for 70,700 plugged wells. A total of 56,039 (44.3%) of the plugged wells did not have reported last production or abandonment dates.

3.3. Proximity to and Risk Ratios for Exposure to Wells

We estimated that, from 2005 to 2009, 395,720 (1.1%) Californians lived within 1 km of new wells, 1,169,930 (3.2%) lived within 1 km of active wells, and 883,394 (2.4%) lived within 1 km of wells that were retired or long-term idle (Table 2). The number and proportion of California residents living near wells decreased during

Table 2

Estimates of the Number of Californians Residing Within 1 km of at Least One Well Stratified by Time Period

	Time period		
	2005-2009	2010-2014	2015-2019
New wells	395,720 (1.1)	214,497 (0.6)	170,679 (0.4)
Active wells	1,169,930 (3.2)	1,150,382 (3.0)	1,109,293 (2.8)
Retired wells	883,394 (2.4)	775,053 (2.0)	588,647 (1.5)
Plugged wells ^a	_	_	8,986,469 (22.9)

Note. We assessed exposure separately for new, active, retired, and (for 2015-2019) plugged wells. The values are reported as number (%) of residents exposed.

^aIncludes wells plugged and abandoned at any time before 2019, including wells retired before the beginning of the study period in 2005.

the study period, with 170,679 (0.4%) living near new wells in 2015–2019, 1,109,293 (2.8%) living near active wells, and 588,647 (1.5%) near wells that were retired or long-term idle. In 2015–2019, nearly 9.0 million (22.9%) California residents lived within 1 km of a well that had been plugged and abandoned. Note that some individuals were exposed to wells in multiple stages (e.g., both active and plugged wells).

3.3.1. Proximity to New Wells

Several marginalized groups had disproportionately high exposure to new wells during the study period. The total number of people exposed to new wells declined by 56.9% through the study period, which correlates with a decline in the number of wells that were drilled or spudding during the study period. The rate of decline was comparatively higher for non-Hispanic white people (66.1%) and lower for Hispanic and Latinx people at 41.6% (Table S2 in Supporting Information S1). Non-Hispanic Black people had the greatest disparities throughout the study period, though there was a steep decline, with RR ranging from 1.90 in 2005–2009 to 1.21 in 2015–2019 (Figure 3).

American Indian and Alaska Native people had disproportionately high exposure in 2010–2014, with an of RR 1.09, declining to 0.62 in 2015–2019. Hispanic and Latinx people also had disproportionately high exposure starting in 2010, with an RR of 1.05 in 2010–2014 increasing to 1.13 in 2015–2019. Non-Hispanic white people



Figure 3. Estimated risk ratios for each racial/ethnic (top row) and socioeconomic group of interest (bottom). We estimated the proportion of each group with any exposure to wells in one of three stages: new wells (left column), active wells (center), and retired wells (right). For each group and well stage, we estimated risk ratios in the three time periods included in the study: 2005–2009, 2010–2014, and 2015–2019. A risk ratio >1 means the group has disproportionately high exposure compared to statewide representation of the group and a risk ratio <1 means the group has disproportionately low exposure, compared to the group's statewide representation.



were not disproportionately exposed at any point and non-Hispanic Asian people had disproportionately low exposure (RR < 1) in all three time periods. Low-income people were more likely to live near new wells in 2005–2009 (RR = 1.12) and 2015–2019 (RR = 1.07), but not in 2010–2014 (RR = 0.96). Renters consistently had high exposure to new wells, with RRs ranging from 1.07 to 1.26. Non-voters had disproportionately high exposure between 2010 and 2019, with an RR of 1.15.

3.3.2. Proximity to Active Wells

We observed similar racial and socioeconomic disparities in residential proximity to active wells. Though there was not a substantial change in the total number of people living near active wells during the study period, the number of non-Hispanic white people exposed decreased by 12.0% and exposed non-Hispanic Black people decreased by 9.4%, while the numbers of exposed non-Hispanic Asian people increased by 15.3% and multiracial people near active wells increased by 38.9% (Table S3 in Supporting Information S1). Despite these changes in absolute number of people exposed to active wells, non-Hispanic Black people were still consistently disproportionately exposed, with an RR of 1.44 in 2005–2009 increasing slightly to 1.49 in 2015–2019. Latinx people also had disproportionately high exposure, with an RR of 1.13 in 2005–2009 declining to 1.04 in 2015–2019. Non-Hispanic American Indian, Alaska Native, Asian, and white people had disproportionately low exposure to active wells throughout the study period (RR < 1). Socioeconomically marginalized people had disproportionately high exposure to active wells throughout the study period. The widest disparities were for renters with RRs between 1.17 and 1.21, non-voters with RRs of 1.15 or 1.16, and low-income people with RRs between 1.12 and 1.21.

3.3.3. Proximity to Retired or Long-Idle Wells

We found similar patterns of exposure disparities for populations near wells that were retired or long-term idle. The number of people living near well retirements decreased by 33.4% during the study period, but non-Hispanic white people had the fastest rate of decline at 43.7% (Table S4 in Supporting Information S1). Non-Hispanic Black people were again consistently disproportionately exposed to retired wells, with an RR of 1.57 in 2005–2009 increasing to 1.68 in 2015-2019. Hispanic or Latinx people also had slight exposure disparities throughout the study period, with RR ranging from 1.04 to 1.07. Once again, American Indian, Alaska Native, non-Hispanic Asian, and non-Hispanic white people either were not disproportionately exposed or had comparatively low exposure. Low-income people, renters, and non-voters were consistently over-represented among populations living near retired wells (RR > 1.1). Linguistically isolated adults had disproportionately high exposure from 2005 to 2014, and people who had not finished high school had higher exposure from 2010 to 2019.

3.3.4. Proximity to Plugged and Abandoned Wells

Finally, there were also similar disparities in exposure to plugged and abandoned wells in 2015–2019, including those plugged before 2005. Non-Hispanic Black people had the greatest exposure disparity (RR = 1.16), followed by Hispanic people (RR = 1.12), non-Hispanic Asian people (RR = 1.04), and those categorized as non-Hispanic other (RR = 1.04). We observed disproportionately low exposure among American Indian and Alaska Native people (RR = 0.68), non-Hispanic white people (RR = 0.85), and non-Hispanic multiracial people (RR = 0.88). By all indicators, socioeconomically marginalized people had disproportionately high exposure to plugged wells in 2015–2019. The widest disparities were for linguistically isolated adults, people who had not finished high school, and non-voters.

3.4. Intensity of Exposure

Among exposed block groups within 1 km of active wells, the average production volume decreased by 28.7% from 670,918 BOE in 2005–2009 to 478,467 in 2015–2019 (Table S5 in Supporting Information S1). There was a wide range in block group-level exposure to cumulative production volume, from a minimum of 10 BOE to a maximum of 14,100,859 BOE in 2005–2009. The number of block groups within 1 km of new wells decreased over the study period, with 1.3% exposed to new wells in 2005–2009 and 0.5% exposed in 2015–2019. The mean number of new wells in exposed block groups increased from 7.2 in 2005–2009 to a peak of 8.8 in 2010–2014, then decreased to 6.5 in 2015–2019. For active wells, 3.4% of block groups were exposed in 2005–2009 and 2.9% were exposed in 2015–2019. Exposed block groups had an average of 22.9 wells nearby in 2005–2009, increasing to an average of 24.3 in 2015–2019. Exposure to well retirements also decreased, from an average of 5.4 wells across 586 exposed block groups in 2005–2009 to an average of 4.1 wells across 367 block groups in 2015–2019.

Among exposed block groups, as exposure to cumulative production volume increased, the percentage of the non-Hispanic Black population also monotonically increased, with the largest exposure disparity in block groups





Figure 4. Estimated risk ratios for exposure to cumulative oil and gas production volume in barrels of oil equivalent stratified by quintile of exposure (from least exposure, 1, to highest exposure, 5), among block groups within 1 km of producing wells. For each racial/ethnic group, we estimated risk ratios by comparing the proportion of the group that was exposed relative to the proportion of the group statewide. A risk ratio >1 means the group has disproportionately high exposure compared to the statewide proportion and a risk ratio <1 means the group has disproportionately low exposure.

with the most intensive production (Figure 4). In block groups in the lowest quintile of exposure to production volume, the RR for non-Hispanic Black people ranged from 0.66 to 1.04 between the three time periods. Exposure disparities for non-Hispanic Black people increased substantially among block groups in the higher exposure quintiles, ranging from 2.05 to 2.39 in the areas with the most intensive production. Hispanic and Latinx people only had disproportionately high exposure in block groups with relatively lower production intensity (i.e., in the first and second quintiles). Non-Hispanic white people had disproportionately low exposure in areas with comparatively less production volume, but not in areas with high production volume. American Indian and Alaska Native people were not disproportionately exposed. These patterns were consistent across all three time

periods. For linguistically isolated people and residents who had not finished high school, observed disparities were higher in areas with less production (Figure S1 in Supporting Information S1). There was not substantial variation in RRs for renters, low-income people, and non-voters.

We also considered exposure to different densities of new and retired wells. As we observed for production volume, non-Hispanic Black people were more likely to live in areas with any drilling and, among those living near new wells, the exposure disparities were widest in areas with more intensive drilling (Figure S2 in Supporting Information S1). However, the pattern was different for well retirements. Non-Hispanic Black people were more likely to live near retired wells, but, except for 2005–2009, the RRs were largest in areas with less intensive activity (Figure S3 in Supporting Information S1).

3.5. County-Level Analyses

More than 90% of Californians residing near new, active, or retired wells during the study period lived in Los Angeles, Orange, or Kern Counties (Table S6 in Supporting Information S1). Los Angeles County had the most exposed residents in all well stages, with 123,314 residents residing near new wells, 725,503 near active wells, 431,051 near retired wells, and 4,663,082 near plugged wells. In Los Angeles County, from 2005 to 2019, non-Hispanic Black people had disproportionately high exposure to oil and gas wells at each stage of production, similar to statewide results (Figure S6 in Supporting Information S1). Hispanic and Latinx people and non-Hispanic white people also had disproportionately high exposure to new wells before 2015, patterns we did not see statewide. Renters were more likely to live near wells at all stages of production. Low-income people had disproportionately high exposure to new wells before 2015, patterns we did not finished high school had lower exposure to new wells. Los Angeles neighborhoods with more intensive oil and gas production also had higher proportions of non-Hispanic Black people (Figure S7 in Supporting Information S1), but not socioeconomically marginalized people (Figure S8 in Supporting Information S1).

In Orange County, non-Hispanic white people had disproportionately high exposure to wells in all development stages from 2005 to 2019 (Figure S9 in Supporting Information S1). American Indian and Alaska Native people living in Orange County had disproportionately high exposure new wells between 2005 and 2014 and to active and retired wells from 2005 to 2009. All other racial/ethnic groups had disproportionately low exposure. There were no substantial differences in these patterns when we examined areas with more intensive production for the racial/ethnic (Figure S10 in Supporting Information S1) and socioeconomically marginalized groups (Figure S11 in Supporting Information S1).

In Kern County, non-Hispanic white people consistently had disproportionately high exposure to wells in all development stages (Figure S12 in Supporting Information S1). American Indian and Alaska Native people had higher exposure to wells in some stages. All other racial/ethnic groups in Kern County had disproportionately low exposure across well stages and throughout the study period. Disparities for non-Hispanic white people were wider in areas with more intensive production, except in 2005–2009 (Figure S13 in Supporting Information S1). Socioeconomically marginalized people were typically less likely to live in neighborhoods near the most productive oil and gas wells (Figure S14 in Supporting Information S1).

3.6. Secondary Analyses

There was not substantial oil and gas development on federally- or state-recognized tribal lands in California. A total 19 wells, all of which were idle or plugged, were located on the lands of the Paskenta Band of Nomlaki Indians (17 wells), the Morongo Band of Mission Indians (1 well), and the Soboba Band of Luiseño Indians (1 well).

In the within block group analyses, we did not observe substantial differences in the racial composition of block groups where the number of new wells changed between 2010–2014 and 2015–2019 (Figure S4 in Supporting Information S1). Non-Hispanic Black people comprised a slightly higher proportion of people in block groups where the count of new wells decreased than in block groups where the count of new wells increased or stayed the same. Non-Hispanic Asian people comprised a higher proportion of people in block groups where the number of new wells did not change. Similarly, we did not observe substantial racial/ethnic differences among block groups where cumulative production volume shifted (Figure S5 in Supporting Information S1). There was slightly higher representation of Latinx and non-Hispanic Black people in block groups where cumulative production volume changed.



4. Discussion

In this longitudinal study, we found that racially and socioeconomically marginalized people had disproportionately high exposure to oil and gas development activities across California between 2005 and 2019. Specifically, the proportion of Black, Hispanic and Latinx, and low-income people residing within 1 km of wells was substantially higher than their representation statewide. For non-Hispanic Black people and people living in poverty, these disparities were persistent throughout the 15-year study period. Individuals living in proximity to wells may be exposed to chemical hazards, noise, and other stressors associated with oil and gas developments (Allshouse et al., 2019; Elliott et al., 2017; Garcia-Gonzales, Shonkoff, et al., 2019). Though oil and gas development activities-and the total number of people exposed to them-declined throughout the study period, the rate of decline was higher for non-Hispanic white people and lower for Hispanic and Latinx people. We observed that while the disparities for non-Hispanic Black people living near new wells declined through the study period, the disparities were consistent for active wells, indicating the persistence of these disparities once wells have been put into production. Among block groups exposed to active oil and gas wells, more intensive production was associated with higher proportion of Black residents, heightening potential health risks faced by these residents. However, we did not observe this pattern for socioeconomically marginalized people. The widest disparities, therefore, were for non-Hispanic Black people in areas of Los Angeles County with the most intensive production, which had 2.4 times the proportion of Black residents compared to the statewide population.

Systemic factors affecting oil and gas-related exposures for Indigenous people may be different than for other racially marginalized groups, as some tribes in some settings have permitted upstream oil and gas production on their tribally-controlled land (Gilio-Whitaker, 2019). We recognize that this may present a different dynamic than in areas where historically marginalized communities have lacked decision-making power over local land uses and where tribal lands have been forcefully seized. In our analysis, we observed fewer than 20 wells on federally- or state-recognized tribal lands in California. However, there may be additional wells located on lands held by tribes that are not federally- or state-recognized and consequently not included in the data set we utilized. Importantly, Farrell et al. (2021) found that the estimated 606,604 square km of tribes' present-day lands are less likely to overlie oil and gas reserves than the over 54 million square km of historical tribal lands, which the authors note has limited tribes' ability to benefit from resource extraction in ways that settlers have.

There were notable differences when we separated out the analysis by county in the three most exposed counties. The results for Los Angeles County mirrored the statewide results, likely because the county is home to most California residents who were exposed to new, active, retired, and plugged wells. Linguistically isolated residents and non-voters had disproportionately low exposure to wells. The results for Orange and Kern Counties were substantially different from Los Angeles and state-wide findings. There was disproportionately high exposure among non-Hispanic white, and American Indian/Alaska Native residents for all well stages, and we did not observe disproportionately high exposure among socioeconomically marginalized residents of either Orange or Kern County. Notably, there are several predominantly Latinx communities in Kern County near active oil fields, some of which have successfully advocated for protections from oil and gas wells in their neighborhoods (Kane, 2020). Collectively, these results indicate that patterns of oil and gas development activities that lead to environmental inequities vary across California.

Land-use decision-making for extractive industries such as oil and gas involve many factors, including geological, economic, and policy considerations, particularly land-use policies at municipal, county, and state-levels (Kaufmann & Cleveland, 2001; Kroepsch et al., 2019). Prior studies indicate that racially and socioeconomically marginalized populations have less capacity to influence local land-use policy and prohibit siting of industrial environmental hazards in their neighborhoods (Ash et al., 2013; Cushing et al., 2015). This is particularly important in communities facing cumulative environmental and social burdens, which may make residents more susceptible to exposure to stressors associated with oil and gas development (National Environmental Justice Advisory Council, 2004). Prior to the study period, historical structural factors appear to have influenced the siting of wells, including racist policies such as redlining that were promulgated by government agencies and private firms, discussed in greater depth below (Cumming, 2018; Gonzalez, Nardone, et al., 2022).

One in five Californians, or nearly 9 million people, lived within 1 km of a plugged and abandoned well in 2015–2019. The racial/ethnic and socioeconomic disparities in exposure to plugged and abandoned wells were similar to disparities observed for new and active wells. This may indicate that these disparities have been

persistent before the study period and for wells that have long been abandoned, though demographic characteristics have likely changed in many areas with sustained oil and gas development. Hazards associated with plugged and abandoned wells are not well understood and likely vary by well type, underlying geology, well age, and other factors (Williams et al., 2021). However, due to data availability and scope, we did not assess these differences. The standards for well plugging and abandonment have varied through time, so wells retired before 1976, when more stringent standards were established, were held to less-stringent requirements, and consequently may pose additional risks. In addition to wells that have been reported as plugged, another way to define postproduction wells include long-term idle (8+ years), which may include unplugged "orphaned" wells (i.e., lacking a legally responsible owner) that must be plugged and abandoned at public expense. Finally, there are an unknown but potentially substantial number of legacy abandoned wells, or those that were drilled prior to consistent record-keeping and which are not included in contemporary datasets, including wells that were never plugged. Methane emissions-often correlated with benzene, toluene, ethylene, and xylene, air toxics that have been associated with numerous adverse health outcomes-have been detected from abandoned wells (Gross et al., 2013; Lebel et al., 2020; McKenzie et al., 2017, 2018; Williams et al., 2021). Prior research has found that methane emissions from unplugged wells are worse than emissions from plugged wells and that gas wells have higher emissions than combined oil and gas wells (Lebel et al., 2020; Williams et al., 2021). Given the widespread potential exposures we observed, more study is needed to verify whether postproduction wells are indeed safe and to determine whether monitoring systems or policy interventions may help mitigate potential hazards.

There is a growing literature investigating associations between living near oil and gas development and adverse health outcomes, and documenting plausible etiologic pathways through which proximity to wells could confer risk (Deziel et al., 2020; Deziel, Clark, et al., 2022; Garcia-Gonzales, Shonkoff, et al., 2019). The majority of epidemiological studies on this topic have reported significant associations (Deziel et al., 2020). Deziel et al. (2020) noted that among 12 studies that examined associations between proximity to oil and gas wells and risk of adverse birth outcomes, most observed positive relationships. Several perinatal studies employed a difference-in-differences study design, which helps control for unmeasured confounding, with each reporting positive associations between exposure to oil and gas development and adverse perinatal outcomes (Currie et al., 2017; Willis, Hill, Boslett, et al., 2021; Willis, Hill, Kile, et al., 2021). Deziel et al. (2020) found suggestive evidence of an association between proximity to oil and gas wells and risk of childhood cancer, though the studies in aggregate are as inconclusive. In a subsequent case-control study, Clark et al. (2022) reported that exposure to unconventional oil and gas development was significantly associated with higher risk of childhood acute lymphoblastic leukemia.

Previously observed health risks may be attributable to multiple physical, chemical, and other stressors associated with oil and gas development (Deziel, Clark, et al., 2022). Contamination of groundwater from oil and gas-related activities can result from spills, leaks from wells, and improper storage of wastewater, also referred to as produced water (Deziel, Clark, et al., 2022; DiGiulio et al., 2021). Noise pollution has been detected near oil and gas operations at levels that may cause sleep disturbance, annoyance, and increase the risk of cardiovascular disease (Allshouse et al., 2019; Blair et al., 2018; Hays et al., 2017). Numerous air pollutants are emitted at each stage of upstream oil and gas development. In a 2019 review, Garcia-Gonzales, Shamasunder, and Jerrett (2019) and Garcia-Gonzales, Shonkoff, et al. (2019) concluded that active oil and gas wells have the greatest potential to emit hazardous air pollutants, including several known carcinogens. Air monitoring in California found that drilling and operating oil and gas wells resulted in elevated levels of ambient air pollutants (including ozone and fine particulates) as far as 4 km downwind and at concentrations that could increase risk of adverse health outcomes (Gonzalez, Francis, et al., 2022; Gonzalez, Nardone, et al., 2022). Taken together, there is substantial evidence that exposure to oil and gas wells can contribute to a range of adverse health outcomes, mediated by air, water, and noise pollution, though additional studies that improve exposure assessment approaches can further elucidate specific etiologic pathways.

There is also prior evidence of disproportionately high exposure to oil and gas infrastructure and related health disparities among racially and socioeconomically marginalized people. A 2014 report found, among the subset of California communities already heavily burdened by pollution according to CalEnviroScreen 2.0, the majority (92%) of people living within 1 mile of wells were of Asian, Black, or Latinx descent (Srebotnjak & Rotkin-Ellman, 2014). Casey et al. (2021) found that California residents of color and those with less political engagement are more likely to live near wells with higher methane emissions. Similar patterns have been observed in other states. Silva et al. (2018) found that injection wells were more likely to be sited in Ohio neighborhoods

with lower income. In a study of exposure to natural gas flaring in Texas, Cushing et al. (2020) found risk of adverse birth outcomes was higher among Hispanic parents compared to non-Hispanic white parents; the authors hypothesize that Hispanic pregnant people may be at higher risk due to disparities in preexisting chronic health conditions, co-exposures to other pollutants, barriers to healthcare access, or other stressors associated with structural racism. Relatedly, two papers by Johnston et al. (2016, 2020) reported that Hispanic and Latinx people had disproportionately high exposure to wastewater disposal wells and natural gas flaring in Texas's Eagle Ford Shale. Clark et al. (2021) found that, though the number of oil and gas-related complaints filed by residents in Pennsylvania did not vary by race/ethnicity, a determination that water supplies were impaired were less likely in areas with higher proportions of racially marginalized people. Proville et al. (2022) found that multiply marginalized populations intersected with more intensive oil and gas development in many areas of the U.S., including California's San Joaquin Valley. In a case-control study, Gonzalez et al. (2020) found that exposure to oil and gas development was associated with higher risk of spontaneous preterm birth among birthing individuals in the San Joaquin Valley, California, and that the risk was confined to births to parents who were Hispanic, Black, or who had completed fewer than 12 years of education. Tran et al., in a set of retrospective cohort studies, found that proximity to active wells with higher oil and gas production volume (2020) and hydraulically fractured wells (2021) were associated with adverse fetal growth outcomes, including small-for-gestational age births, low birthweight births and lower term birth weight. In both studies, Tran et al. adjusted for known individualand neighborhood-level potential confounders identified a priori, and restricted their analyses to individuals within 10 km wells to reduce potential confounding from unobserved covariates. Also in both studies, Tran et al. observed strongest effects among rural residents. Notably, some studies that have reported racial and socioeconomic disparities have been cross-sectional and may be subject to residual confounding.

Possible mechanisms leading to observed disparities in exposure to oil and gas development and associated health outcomes emerge from regulatory and land-use decision-making processes involving multiple stakeholders (Kroepsch et al., 2019). In the current study, we observed distributive injustice, with disproportionately high exposure among racial and socioeconomically marginalized people. Prior studies have examined the role of procedural injustice, including a racist policy known as redlining that, starting in the 1930s, designated neighborhoods as hazardous or high-risk for investment or loans (Hillier, 2003, 2005; Swope et al., 2022). The real estate and banking industries, along with government agencies like the federal Home Owners Loan Corporation (HOLC), appraised neighborhood-level investment risk and systematically assigned poorer grades to neighborhoods with Black, immigrant, and other marginalized groups (Hillier, 2003, 2005; Swope et al., 2022). In a case study of Los Angeles, Cumming (2018) describes how the racial composition of neighborhoods affected how HOLC staff graded neighborhoods, finding that in some predominantly white neighborhoods with oil wells, residents secured better grades by negotiating with HOLC staff. In other neighborhoods, the presence of both oil wells and people of color were cited as justification for assigning worse grades. Gonzalez, Nardone, et al. (2022) found that historical redlining was associated with the siting of oil and gas wells in 33 cities across the U.S., including Los Angeles. Neighborhoods with worse HOLC grades had significantly more wells, with the highest number and density of wells in historically redlined neighborhoods (Gonzalez, Nardone, et al., 2022). Historical redlining and other local, state, and federal policies that engendered and perpetuated racial segregation may partially account for the persistent racial disparities observed in the current study.

Our study had several limitations. Sociodemographic data are not publicly available at the individual level, so we were not able to identify individuals with intersecting marginalized identities. The U.S. Census Bureau provides data on some intersecting marginalized identities, and investigators should consider leveraging these data in future research. Because we assessed exposure at the area level, rather than the individual household, there is potential for exposure misclassification where, for example, residences are clustered on one side of a block group. This could be improved in future studies by determining where residences are located within census block groups, particularly for larger block groups (i.e., dasymetric mapping). The under-reporting of legacy abandoned wells likely resulted in an underestimate of the populations exposed to these wells. Lebel et al. (2020) estimated that abandoned wells in California have been under-counted by approximately 17%, based on historical maps from the U.S. Geological Survey. Williams et al. (2021) estimated that there are approximately 200,000 abandoned wells in California, which is roughly 74,000 or 58% more plugged and abandoned wells than were reported in the CalGEM data set we used in the current study. Due to insufficient plugging (or no plugging at all), these wells may emit air toxics that can be harmful to health, though to date there is only limited evidence of these potential hazards (Williams et al., 2021). More work is needed to identify locations of these wells and,

importantly, whether insufficiently plugged wells indeed pose environmental health or safety risks for nearby residents. Additionally, examining exposure to other oil and gas infrastructure, such as pipelines and refineries, was outside our scope. However, due to potential risks associated with exposure to other infrastructure, more work assessing exposure to these potential hazards could help determine whether environmental inequities exist. There was also substantial missingness in last date of production or abandonment date. Some 44.3% of plugged wells had no reported date of last production or abandonment. Due to the long duration of oil and gas production in California, and the paucity of records from earlier periods, it is likely that many of these plugged wells were abandoned prior to 2005. Based on this, we do not expect there to be substantial misclassification of exposure to wells that were retired during the study period. In a secondary analysis, we assessed exposure to all wells labeled as plugged, and similarly observed that racially marginalized people were more likely to live near plugged wells regardless of whether an abandonment date was reported. Idle wells with a last date of production but no abandonment date may become active again, though the 8-year period we used to define long-idle abandoned wells accords with the state's definition of abandoned wells (California Code PRC § 3008).

We investigated whether racially and socioeconomically marginalized people in California were more likely to live in neighborhoods with oil and gas wells in all stages of production. Black, Latinx, and socioeconomically marginalized people had disproportionately high exposure to new, active, retired, and plugged oil and gas wells between 2005 and 2019. Black people not only had disproportionately high exposure to any wells; when they were exposed, they were more likely to be in areas with the most intensive production. Exposure to wells at each stage of development presents risks to health, and health disparities reported in prior studies may be partly attributable to disparities in exposure to wells. Across California, state and local policymakers have considered options to provide protections for frontline communities near oil and gas development facilities. Potential policy interventions aimed at mitigating risks associated with oil and gas development include on-site engineering controls, setbacks or buffer distances between wells and sensitive receptors (e.g., residences, schools, medical facilities), and the elimination of oil and gas production (Deziel, McKenzie, et al., 2022). In September 2022, the California legislature passed a new statute that requires a 1 km setback between new wells and sensitive receptors, including residences (Senate Bill No. 1137, 2022). Jurisdictions such as Los Angeles and Culver City have banned new drilling, and Ventura County and the city of Arvin have instituted setback policies and restrictions on drilling and production (Kane, 2020). In forging potential interventions, policymakers should also address environmental health disparities. Further study could expand knowledge on hazards associated with living near oil and gas wells and mechanisms that have led to observed racial and socioeconomic health disparities.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

Sociodemographic data used in the analyses presented here were obtained from the publicly available ACS conducted by the U.S. Census Bureau, specifically, 5-year estimates for 2005–2009, 2010–2014, and 2015–2019. We obtained publicly available data on oil and gas development from the California Geologic Energy Management Division (CalGEM). Additional data came from the Enverus, a private oil and gas data aggregation service; academic researchers may request access to these data for research purposes. All analytic work and visualizations presented in this paper were done using R v. 4.2 software (R Core Team, 2022) as well as the "tidyverse" and "sf" packages (Pebesma, 2018; Wickham et al., 2019). The codebase and processed data set used in these analyses are licensed under MIT, and the GitHub repostory (v1.0.3) with these resources is availble on the FAIR-compliant Zenodo platform at https://doi.org/10.5281/zenodo.7747408 (Gonzalez & Morton, 2023).

References

Adgate, J. L., Goldstein, B. D., & McKenzie, L. M. (2014). Potential public health hazards, exposures and health effects from unconventional natural gas development. *Environmental Science & Technology*, *48*(15), 8307–8320. https://doi.org/10.1021/es404621d

Allshouse, W. B., McKenzie, L. M., Barton, K., Brindley, S., & Adgate, J. L. (2019). Community noise and air pollution exposure during the development of a multi-well oil and gas pad. *Environmental Science & Technology*, 53(12), 7126–7135. https://doi.org/10.1021/acs.est.9b00052
Ash, M., Boyce, J. K., Chang, G., & Scharber, H. (2013). Is environmental justice good for white folks? Industrial air toxics exposure in urban America. *Social Science Quarterly*, 94(3), 616–636. https://doi.org/10.1111/j.1540-6237.2012.00874.x

Acknowledgments

This study was supported by the California Air Resources Board (#18RD018) and National Institute of Environmental Health Sciences (R00 ES027023).

- Blair, B. D., Brindley, S., Dinkeloo, E., McKenzie, L. M., & Adgate, J. L. (2018). Residential noise from nearby oil and gas well construction and drilling. *Journal of Exposure Science and Environmental Epidemiology*, 28(6), 538–547. https://doi.org/10.1038/s41370-018-0039-8 California Tribal Court-State Court Forum. (n.d.). California tribal communities. Retrieved from https://www.courts.ca.gov/3066.htm
- Caron-Beaudoin, É., Bouchard, M., Wendling, G., Barroso, A., Bouchard, M. F., Ayotte, P., et al. (2019). Urinary and hair concentrations of trace metals in pregnant women from Northeastern British Columbia, Canada: A pilot study. *Journal of Exposure Science and Environmental Epidemiology*, 29(5), 613–623. https://doi.org/10.1038/s41370-019-0144-3
- Caron-Beaudoin, É., Valter, N., Chevrier, J., Ayotte, P., Frohlich, K., & Verner, M.-A. (2018). Gestational exposure to volatile organic compounds (VOCs) in Northeastern British Columbia, Canada: A pilot study. *Environment International*, 110, 131–138. https://doi.org/10.1016/j. envint.2017.10.022
- Caron-Beaudoin, É., Whyte, K. P., Bouchard, M. F., Chevrier, J., Haddad, S., Copes, R., et al. (2022). Volatile organic compounds (VOCs) in indoor air and tap water samples in residences of pregnant women living in an area of unconventional natural gas operations: Findings from the EXPERIVA study. Science of the Total Environment, 805, 150242. https://doi.org/10.1016/j.scitotenv.2021.150242
- Casey, J. A., Cushing, L., Depsky, N., & Morello-Frosch, R. (2021). Climate justice and California's methane superemitters: Environmental equity assessment of community proximity and exposure intensity. *Environmental Science & Technology*, 55(21), 14746–14757. https://doi. org/10.1021/acs.est.1c04328
- Casey, J. A., Goin, D. E., Rudolph, K. E., Schwartz, B. S., Mercer, D., Elser, H., et al. (2019). Unconventional natural gas development and adverse birth outcomes in Pennsylvania: The potential mediating role of antenatal anxiety and depression. *Environmental Research*, 177, 108598. https://doi.org/10.1016/j.envres.2019.108598
- Clark, C. J., Johnson, N. P., Soriano, M., Jr., Warren, J. L., Sorrentino, K. M., Kadan-Lottick, N. S., et al. (2022). Unconventional oil and gas development exposure and risk of childhood acute lymphoblastic leukemia: A case-control study in Pennsylvania, 2009–2017. Environmental Health Perspectives, 130(8), 87001. https://doi.org/10.1289/EHP11092
- Clark, C. J., Warren, J. L., Kadan-Lottick, N., Ma, X., Bell, M. L., Saiers, J. E., & Deziel, N. C. (2021). Community concern and government response: Identifying socio-economic and demographic predictors of oil and gas complaints and drinking water impairments in Pennsylvania. *Energy Research & Social Science*, 76, 102070. https://doi.org/10.1016/j.erss.2021.102070
- Cumming, D. G. (2018). Black Gold, white power: Mapping oil, real estate, and racial segregation in the Los Angeles basin, 1900–1939. Engaging Science, Technology, and Society, 4, 85–110. https://doi.org/10.17351/ests2018.212
- Currie, J., Greenstone, M., & Meckel, K. (2017). Hydraulic fracturing and infant health: New evidence from Pennsylvania. *Science Advances*, 3(12), e1603021. https://doi.org/10.1126/sciadv.1603021
- Cushing, L., Morello-Frosch, R., Wander, M., & Pastor, M. (2015). The haves, the have-nots, and the health of everyone: The relationship between social inequality and environmental quality. *Annual Review of Public Health*, 36(1), 193–209. https://doi.org/10.1146/ annurev-publhealth-031914-122646
- Cushing, L. J., Vavra-Musser, K., Chau, K., Franklin, M., & Johnston, J. E. (2020). Flaring from unconventional oil and gas development and birth outcomes in the Eagle Ford Shale in South Texas. *Environmental Health Perspectives*, 128(7), 77003. https://doi.org/10.1289/ehp6394
- Czolowski, E. D., Santoro, R. L., Srebotnjak, T., & Shonkoff, S. B. C. (2017). Toward consistent methodology to quantify populations in proximity to oil and gas development: A national spatial analysis and review. *Environmental Health Perspectives*, 125(8), 086004. https://doi. org/10.1289/ehp1535
- Deziel, N. C., Brokovich, E., Grotto, I., Clark, C. J., Barnett-Itzhaki, Z., Broday, D., & Agay-Shay, K. (2020). Unconventional oil and gas development and health outcomes: A scoping review of the epidemiological research. *Environmental Research*, 182, 109124. https://doi.org/10.1016/j. envres.2020.109124
- Deziel, N. C., Clark, C. J., Casey, J. A., Bell, M. L., Plata, D. L., & Saiers, J. E. (2022). Assessing exposure to unconventional oil and gas development: Strengths, challenges, and implications for epidemiologic research. *Current Environmental Health Reports*, 9(3), 436–450. https:// doi.org/10.1007/s40572-022-00358-4
- Deziel, N. C., McKenzie, L. M., Casey, J. A., McKone, T. E., Johnston, J. E., Gonzalez, D. J. X., et al. (2022). Applying the hierarchy of controls to oil and gas development. *Environmental Research Letters: ERL [Web Site]*, 17(7), 071003. https://doi.org/10.1088/1748-9326/ac7967
- DiGiulio, D., & Shonkoff, S. B. C. (2021). Impact to groundwater resources from disposal of produced water into unlined produced water ponds in the San Joaquin Valley. In A Preliminary Evaluation of Data Collected Pursuant to California Senate Bill 1281: An Independent Review of Scientific and Technical Information. California Council on Science and Technology. Retrieved from https://ccst.us/reports/ oil-and-gas-water-cycle-reporting/
- DiGiulio, D. C., Rossi, R. J., Jaeger, J. M., Shonkoff, S. B. C., & Ryan, J. N. (2021). Vulnerability of groundwater resources underlying unlined produced water ponds in the Tulare Basin of the San Joaquin Valley, California. *Environmental Science & Technology*, 55(21), 14782–14794. https://doi.org/10.1021/acs.est.1c02056
- Elliott, E. G., Ettinger, A. S., Leaderer, B. P., Bracken, M. B., & Deziel, N. C. (2017). A systematic evaluation of chemicals in hydraulic-fracturing fluids and wastewater for reproductive and developmental toxicity. *Journal of Exposure Science and Environmental Epidemiology*, 27(1), 90–99. https://doi.org/10.1038/jes.2015.81
- Ennis, S. R., Rios-Vargas, M., & Albert, N. G. (2011). *The Hispanic population: 2010 Census Briefs (No. C2010BR-04)*. U.S. Census Bureau. Farrell, J., Burow, P. B., McConnell, K., Bayham, J., Whyte, K., & Koss, G. (2021). Effects of land dispossession and forced migration on Indig-
- enous peoples in North America. *Science*, *374*(6567), eabe4943. https://doi.org/10.1126/science.abe4943 Ferrar, K. (2020). Systematic racism in Kern County oil and gas permitting ordinance. Retrieved from https://www.fractracker.org/2020/06/ systematic-racism-in-kern-county-oil-and-gas-permitting-ordinance/
- Garcia-Gonzales, D. A., Shamasunder, B., & Jerrett, M. (2019). Distance decay gradients in hazardous air pollution concentrations around oil and natural gas facilities in the city of Los Angeles: A pilot study. *Environmental Research*, 173, 232–236. https://doi.org/10.1016/j. envres.2019.03.027
- Garcia-Gonzales, D. A., Shonkoff, S. B. C., Hays, J., & Jerrett, M. (2019). Hazardous air pollutants associated with upstream oil and natural gas development: A critical synthesis of current peer-reviewed literature. *Annual Review of Public Health*, 40(1), 283–304. https://doi.org/10.1146/ annurev-publhealth-040218-043715
- Gilio-Whitaker, D. (2019). As long as grass grows: The indigenous fight for environmental justice, from colonization to standing rock. Beacon Press.
- Gonzalez, D. J. X., Francis, C. K., Shaw, G. M., Cullen, M. R., Baiocchi, M., & Burke, M. (2022). Upstream oil and gas production and ambient air pollution in California. *The Science of the Total Environment*, 806, 150298. https://doi.org/10.1016/j.scitotenv.2021.150298
- Gonzalez, D. J. X., & Morton, C. M. (2023). Temporal trends of racial and socioeconomic disparities in population exposures to upstream oil and gas development in California: Codebase and Dataset. [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7587120

- Gonzalez, D. J. X., Nardone, A., Nguyen, A. V., Morello-Frosch, R., & Casey, J. A. (2022). Historic redlining and the siting of oil and gas wells in the United States. *Journal of Exposure Science and Environmental Epidemiology*, 33(1), 76–83. https://doi.org/10.1038/s41370-022-00434-9 Gonzalez, D. J. X., Sherris, A. R., Yang, W., Stevenson, D. K., Padula, A. M., Baiocchi, M., et al. (2020). Oil and gas production and sponta-
- neous preterm birth in the San Joaquin Valley, CA: A case-control study. *Environmental Epidemiology*, 4(4), e099. https://doi.org/10.1097/ee9.000000000000099
- Gross, S. A., Avens, H. J., Banducci, A. M., Sahmel, J., Panko, J. M., & Tvermoes, B. E. (2013). Analysis of BTEX groundwater concentrations from surface spills associated with hydraulic fracturing operations. *Journal of the Air & Waste Management Association*, 63(4), 424–432. https://doi.org/10.1080/10962247.2012.759166
- Hays, J., McCawley, M., & Shonkoff, S. B. C. (2017). Public health implications of environmental noise associated with unconventional oil and gas development. *Science of the Total Environment*, 580, 448–456. https://doi.org/10.1016/j.scitotenv.2016.11.118
- Hill, A. B. (1965). The environment and disease: Association or causation? Proceedings of the Royal Society of Medicine, 58(5), 295–300. https:// doi.org/10.1177/003591576505800503
- Hillier, A. E. (2003). Redlining and the Home Owners' Loan Corporation. Journal of Urban History, 29(4), 394–420. https://doi. org/10.1177/0096144203029004002
- Hillier, A. E. (2005). Residential security maps and neighborhood appraisals: The home owners' loan corporation and the case of Philadelphia. Social Science History, 29(2), 207–233. https://doi.org/10.1017/s014555320001292x
- Hoeffel, E. M., Rastogi, S., Kim, M. O., & Shahid, H. (2012). The Asian population: 2010 census Briefs (No. C2010BR-11). U.S. Census Bureau. Johnston, J. E., Chau, K., Franklin, M., & Cushing, L. (2020). Environmental justice dimensions of oil and gas flaring in South Texas: Disproportionate exposure among Hispanic communities. Environmental Science & Technology, 54(10), 6289–6298. https://doi.org/10.1021/acs. est 0c00410
- Johnston, J. E., Enebish, T., Eckel, S. P., Navarro, S., & Shamasunder, B. (2021). Respiratory health, pulmonary function and local engagement in urban communities near oil development. *Environmental Research*, 197, 111088. https://doi.org/10.1016/j.envres.2021.111088
- Johnston, J. E., Werder, E., & Sebastian, D. (2016). Wastewater disposal wells, fracking, and environmental injustice in southern Texas. American Journal of Public Health, 106(3), 550–556. https://doi.org/10.2105/ajph.2015.303000
- Kane, J. (2020). Tired of wells that threaten residents' health, a small California town takes on the oil industry. Inside Climate News. Retrieved from https://insideclimatenews.org/news/03082020/california-big-oil-environmental-health/
- Kaufmann, R. K., & Cleveland, C. J. (2001). Oil production in the lower 48 states: Economic, geological, and institutional determinants. *Energy Journal*, 22(1), 27–49. https://doi.org/10.5547/issn0195-6574-ej-vol22-no1-2
- Kroepsch, A. C., Maniloff, P. T., Adgate, J. L., McKenzie, L. M., & Dickinson, K. L. (2019). Environmental justice in unconventional oil and natural gas drilling and production: A critical review and research agenda. *Environmental Science & Technology*, 53(12), 6601–6615. https:// doi.org/10.1021/acs.est.9b00209
- Lebel, E. D., Lu, H. S., Vielstädte, L., Kang, M., Banner, P., Fischer, M. L., & Jackson, R. B. (2020). Methane emissions from abandoned oil and gas wells in California. *Environmental Science & Technology*, 54(22), 14617–14626. https://doi.org/10.1021/acs.est.0c05279
- Lucas, R. M., & McMichael, A. J. (2005). Association or causation: Evaluating links between "environment and disease". *Bulletin of the World Health Organization*, 83(10), 792–795.
- McKenzie, L. M., Allshouse, W. B., Byers, T. E., Bedrick, E. J., Serdar, B., & Adgate, J. L. (2017). Childhood hematologic cancer and residential proximity to oil and gas development. *PLoS One*, 12(2), e0170423. https://doi.org/10.1371/journal.pone.0170423
- McKenzie, L. M., Blair, B., Hughes, J., Allshouse, W. B., Blake, N. J., Helmig, D., et al. (2018). Ambient nonmethane hydrocarbon levels along Colorado's Northern Front Range: Acute and chronic health risks. *Environmental Science & Technology*, 52(8), 4514–4525. https://doi. org/10.1021/acs.est.7b05983
- Michanowicz, D. R., Williams, S. R., Buonocore, J. J., Rowland, S. T., Konschnik, K. E., Goho, S. A., & Bernstein, A. S. (2019). Population allocation at the housing unit level: Estimates around underground natural gas storage wells in PA, OH, NY, WV, MI, and CA. *Environmental Health: A Global Access Science Source*, 18(1), 58. https://doi.org/10.1186/s12940-019-0497-z
- National Environmental Justice Advisory Council. (2004). Ensuring risk reduction in communities with multiple stressors: Environmental justice and cumulative risks/impacts. National Environmental Justice Advisory Council.
- Pebesma, E. (2018). Simple features for R: Standardized support for spatial vector data. The R Journal, 10(1), 439. https://doi.org/10.32614/ RJ-2018-009
- Proville, J., Roberts, K. A., Peltz, A., Watkins, L., Trask, E., & Wiersma, D. (2022). The demographic characteristics of populations living near oil and gas wells in the USA. *Population and Environment*, 44(1–2), 1–14. https://doi.org/10.1007/s11111-022-00403-2
- Quam-Wickham, N. (1998). "Cities sacrificed on the altar of oil": Popular opposition to oil development in 1920s Los Angeles. Environmental History, 3(2), 189–209. https://doi.org/10.2307/3985379
- R Core Team. (2022). R: A language and environment for statistical computing. R Foundation for Statistical Computing. Retrieved from https://www.R-project.org/
- Senate Bill No. 1137. (2022). Oil and gas: Operations: Location restrictions: Notice of intention: Health protection zone: Sensitive receptors. Public Law No. Senate Bill No. 1137. Retrieved from https://leginfo.legislature.ca.gov/faces/billNavClient.xhtml?bill_id=202120220SB1137
- Shamasunder, B., Collier-Oxandale, A., Blickley, J., Sadd, J., Chan, M., Navarro, S., et al. (2018). Community-based health and exposure study around urban oil developments in south Los Angeles. *International Journal of Environmental Research and Public Health*, 15(1), 138. https://doi.org/10.3390/ijerph15010138
- Shonkoff, S. B. C., Hill, L. L., & Domen, J. K. (2021). California produced water quality: Implications for human health and the environment. In A Preliminary Evaluation of Data Collected Pursuant to California Senate Bill 1281, An Independent Review of Scientific and Technical Information. California Council on Science and Technology. Retrieved from https://ccst.us/reports/oil-and-gas-water-cycle-reporting/
- Shonkoff, S. B. C., Jordan, P., Brandt, A., Ferrar, K., Maddalena, R. L., Greenfield, B. K., et al. (2015). Public health risks associated with current oil and gas development in the Los Angeles Basin. In An Independent Scientific Assessment of Well Stimulation in California. California Council on Science and Technology. Retrieved from http://ccst.us/publications/2015/vol-III-chapter-4.pdf
- Shonkoff, S. B. C., Maddalena, R. L., Hays, J., Stringfellow, W., Wettstein, Z. S., Harrison, R., et al. (2015). Potential impacts of well stimulation on human health in California. In An independent scientific assessment of well stimulation in California. California Council on Science and Technology. Retrieved from http://ccst.us/publications/2015/vol-II-chapter-6.pdf
- Shonkoff, S. B. C., Morello-Frosch, R., Casey, J. A., Deziel, N., DiGiulio, D. C., Foster, S., et al. (2021). Response to CalGEM Questions for the California Oil and Gas Public Health Rulemaking Scientific Advisory Panel. Retrieved from https://www.gov.ca.gov/wp-content/ uploads/2021/10/Public-Health-Panel-Memo.pdf
- Silva, G. S., Warren, J. L., & Deziel, N. C. (2018). Spatial modeling to identify sociodemographic predictors of hydraulic fracturing wastewater injection wells in Ohio census block groups. *Environmental Health Perspectives*, 126(6), 067008. https://doi.org/10.1289/ehp2663



- Solis, N. (2022). California oil regulator confirms methane leak at idle oil wells in Bakersfield. Los Angeles Times. Retrieved from https://www. latimes.com/california/story/2022-05-22/california-oil-regulator-reports-methane-leak-at-idle-oil-well-in-bakersfield
- Srebotnjak, T., & Rotkin-Ellman, M. (2014). Drilling in California: Who's at risk? National Resources Defense Council.
- Su, J. G., Morello-Frosch, R., Jesdale, B. M., Kyle, A. D., Shamasunder, B., & Jerrett, M. (2009). An index for assessing demographic inequalities in cumulative environmental hazards with application to Los Angeles, California. *Environmental Science & Technology*, 43(20), 7626–7634. https://doi.org/10.1021/es901041p
- Swope, C. B., Hernández, D., & Cushing, L. J. (2022). The relationship of historical redlining with present-day neighborhood environmental and health outcomes: A scoping review and conceptual model. *Journal of Urban Health: Bulletin of the New York Academy of Medicine*, 99(6), 959–983. https://doi.org/10.1007/s11524-022-00665-z
- Tran, K. V., Casey, J. A., Cushing, L. J., & Morello-Frosch, R. (2020). Residential proximity to oil and gas development and birth outcomes in California: A retrospective cohort study of 2006–2015 births. *Environmental Health Perspectives*, 128(6), 067001. https://doi.org/10.1289/ ehp5842
- Tran, K. V., Casey, J. A., Cushing, L. J., & Morello-Frosch, R. (2021). Residential proximity to hydraulically fractured oil and gas wells and adverse birth outcomes in urban and rural communities in California (2006–2015). *Environmental Epidemiology*, 5(6), e172. https://doi. org/10.1097/ee9.000000000000172
- U.S. Census Bureau. (2018). TIGER/Line shapefiles and TIGER/line files technical documentation: Chapter 3-geographic shapefile concepts overview. [Dataset]. U.S. Census Bureau. Retrieved from https://www2.census.gov/geo/pdfs/maps-data/data/tiger/tgrshp2018/TGRSHP2018_ TechDoc_Ch3.pdf
- U.S. Census Bureau. (2022). American Community Survey 5-year data (2009–2021). [Dataset]. U.S. Census Bureau. Retrieved from https:// www.census.gov/data/developers/data-sets/acs-5year.html
- U.S. Census Bureau. (2022). Geographic glossary. [Dataset]. U.S. Census Bureau. Retrieved from https://www.census.gov/programs-surveys/ geography/about/glossary.html
- Wickham, H., Averick, M., Bryan, J., Chang, W., McGowan, L. D., François, R., et al. (2019). Welcome to the tidyverse. *Journal of Open Source Software*, 4(43), 1686. https://doi.org/10.21105/joss.01686
- Williams, J. P., Regehr, A., & Kang, M. (2021). Methane emissions from abandoned oil and gas wells in Canada and the United States. Environmental Science & Technology, 55(1), 563–570. https://doi.org/10.1021/acs.est.0c04265
- Willis, M. D., Hill, E. L., Boslett, A., Kile, M. L., Carozza, S. E., & Hystad, P. (2021). Associations between residential proximity to oil and gas drilling and term birth weight and small-for-gestational-age infants in Texas: A difference-in-differences analysis. *Environmental Health Perspectives*, 129(7), 077002. https://doi.org/10.1289/ehp7678
- Willis, M. D., Hill, E. L., Kile, M. L., Carozza, S., & Hystad, P. (2021). Associations between residential proximity to oil and gas extraction and hypertensive conditions during pregnancy: A difference-in-differences analysis in Texas, 1996–2009. International Journal of Epidemiology, 51(2), 525–536. https://doi.org/10.1093/ije/dyab246